

## Alternative reproductive tactics and evolutionary rescue

Robert J. Knell<sup>1</sup> and Jonathan M. Parrett<sup>2</sup>

<sup>1</sup>School of Natural Sciences, University of Hull, Hull, United Kingdom <sup>2</sup>Evolutionary Biology Group, Faculty of Biology, Adam Mickiewicz University, Poznań, Poland

Corresponding author: Robert J. Knell, Hardy Building, The University of Hull, Kingston upon Hull, The East Riding of Yorkshire, HU6 7RX, United Kingdom. Email: rj.knell@hull.ac.uk

#### Abstract

Almost all life on earth is facing environmental change, and understanding how populations will respond to these changes is of urgent importance. One factor that is known to affect the speed by which a population can evolve when faced with changes in the environment is strong sexual selection. This increases the adaptive capacity of a population by increasing reproductive skew toward well-adapted (usually) males who will, on average, be best able to compete for matings. This effect could potentially be disrupted when males pursue alternative reproductive tactics (ARTs), whereby males within a species exhibit qualitatively different behaviors in their pursuit of matings. ARTs are diverse, but one common class is those expressed through condition-dependent polyphenism such that high-quality, well-adapted males compete aggressively for mates and low-quality, poorly adapted males attempt to acquire matings via other, nonaggressive behaviors. Here, using an individual-based modeling approach, we consider the possible impacts of ARTs on adaptation and evolutionary rescue. When the ART is simultaneous, meaning that low-quality males not only engage in contests but also pursue other tactics, adaptive capacity is reduced and evolutionary rescue, where a population avoids extinction by adapting to a changing environment, becomes less likely. This is because the use of the ART allows low-quality males to contribute more maladaptive genes to the population than would happen otherwise. When the ART is fixed, however, such that low-quality males will only use the alternative tactic and do not engage in contests, we find the opposite: adaptation happens more quickly and evolutionary rescue when the environment changes is more likely. This surprising effect is caused by an increase in the mating success of the highest quality males who face many fewer competitors in this scenario-counterintuitively, the presence of males pursuing the ART increases reproductive skew toward those males in the best condition.

Keywords: adaptation, changing environments, "sneak" matings, polyphenism, sexual selection, mating systems

#### Lay Summary

When the environment is changing, organisms can either migrate, go extinct, or become adapted to the new environment—this last process is called "evolutionary rescue." Whether evolutionary rescue happens is strongly affected how mating occurs in the species, with extinction being less likely when mating is like that found in birds of paradise or red deer, where males compete strongly to mate with females. Evolutionary rescue occurs in these systems because the successful males tend to be in good condition and able to cope with the new environment and their "good quality" genes get transmitted to more of the next generation. In many of these species, however, there are some males who "cheat": These are often low-quality, poorly adapted males who would not be successful if they competed normally, so they try to gain matings by other means by, e.g., sneaking past males guarding females or by trying to mate with females on their way to the places where other males compete. We asked how these "alternative reproductive tactics" (ARTs) could change evolutionary rescue by using a simulation model of populations evolving under environmental change. We found that if males using the ART also compete normally for females, the extinction risk is increased because the low-quality, poorly adapted males are better able to get matings and father offspring. When the ART is "fixed," however, with poor-quality males only able to use the alternative tactic and not engaging in contests, we found that the presence of males following the ART decreased the risk of extinction. This surprising result is because when these males exclude themselves from the contests for females, the males with the best, most adapted genes father even more offspring than they would do otherwise, more than counterbalancing the input of low-quality genes from the males pursuing the ART.

### Introduction

Organisms usually exist in habitats that they are reasonably well adapted to. When the environment changes such that those organisms suffer substantial reductions in fitness, there are several possible outcomes: The population in question might become extinct, or the organisms might migrate to or otherwise colonize new environments that have become more suitable to them. A third possibility is "evolutionary rescue": the

phenomenon whereby the affected population evolves rapidly enough to become adapted to the new environment and thus is able to persist despite the environmental changes (Bell, 2017). Evolutionary rescue will only occur when the rate of adaptation is sufficient to offset the fitness reductions arising from environmental change. Understanding the factors that determine how fast adaptation can occur is therefore necessary for predicting responses to changing environments. Given that the environment

Received July 4, 2023; revisions received February 16, 2024; accepted February 24, 2024

<sup>©</sup> The Author(s) 2024. Published by Oxford University Press on behalf of The Society for the Study of Evolution (SSE) and European Society for Evolutionary Biology (ESEN).

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

is changing rapidly in almost every habitat on the planet, and many populations have a limited ability to migrate due to geographical barriers or habitat fragmentation, this is now of considerable importance.

Over recent decades, it has become clear that mating systems can be important determinants of adaptive capacity for animal populations-there is now firm evidence from laboratory studies that strong sexual selection can enhance adaptation and therefore the probability of evolutionary rescue (Cally et al., 2019; Godwin et al., 2020; Jarzebowska & Radwan, 2010; Parrett & Knell, 2018; see Parrett et al., 2019 for a field example). These positive effects of strong sexual selection are a consequence of sexual signals expressed by (usually) males being condition dependent (Dougherty, 2021; Rowe & Houle, 1996; Tomkins et al., 2004), meaning that they respond disproportionately to the overall health and well-being of the bearer. If females base their mating decisions on these signals, or if male-male contests are determined on the basis of their expression, then males that are well adapted to the new environment, and therefore in better condition, will gain a high proportion of matings and father a higher proportion of the offspring in the next generation than would otherwise occur, leading to more rapid adaptation and potentially a greater probability of evolutionary rescue (Agrawal & Whitlock, 2012; Lorch et al., 2003; Martínez-Ruiz & Knell, 2017; Whitlock, 2000).

Mating systems are, of course, greatly diverse and strong sexual selection with either female-choice or intrasexual contests, leading to high reproductive skew in favor of a few high-quality males is only one of many possible variants, each of which could have differing effects on the rate of adaptation and the potential for evolutionary rescue. Here we consider the potential role of one widespread variant, whereby males adopt alternative reproductive tactics (ARTs); either competing with rivals for access to mates (variously called "major," "fighter," "bourgeois," or "guarder" males) or attempting to acquire matings by means other than direct competition ("minor," "scrambler," "parasitic," or "sneak" males) (Engqvist & Taborsky, 2016; Gross, 1996; Neff & Svensson, 2013; Oliveira et al., 2008). ARTs are diverse and distributed across the animal kingdom. Some, such as those found in the ruff, Philomachus pugnax (Lamichhaney et al., 2016; Lank et al., 1995), and the side-blotched lizard, Uta stansburiana (Sinervo & Lively, 1996), are largely determined by genetic polymorphisms associated with specific mating phenotypes, with the polymorphism often being maintained by frequencydependent selection. Others, however, are polyphenisms, whereby any single genotype can develop into multiple phenotypes. In the dung beetle Onthophagus taurus, e.g., males either develop into horned "major" males or effectively hornless "minor" males depending on food availability when larvae (Hunt & Simmons, 1997; Moczek & Emlen, 1999). The horned major males guard females in tunnels and will fight other major males who intrude, but minor males do not fight and instead attempt to acquire matings using nonaggressive tactics including mating with females when they leave the burrow to collect dung or passing a guarding male without fighting to access the female beetle (Moczek & Emlen, 2000).

In general, the polyphenic expression of morphs, which engage in alternative tactics, is thought to be an example of conditional expression, whereby individuals that are small, weak, or otherwise of low status adopt alternative, nonaggressive mating tactics because these tactics lead to higher fitness for them. Large, strong, or otherwise high-status males, on the other hand, gain the greatest fitness benefits from competing aggressively (Gross, 1996; Hunt & Simmons, 2001; Tomkins & Hazel, 2007). Since the generally accepted explanation for the mechanism by which sexual selection enhances adaptation is based on the idea that males in good condition (high status) acquire the majority of matings, it might be expected that this effect should be disrupted by the expression of ARTs since these increase the mating success of males in poor condition, weakening selection against deleterious alleles or for beneficial ones.

This potential effect of ARTs on adaptation is, of course, dependent on ART expression having a heritable component such that those individuals expressing the ART are indeed poorly adapted or carrying a high mutational load when compared with those individuals pursuing the more competitive reproductive tactic. The genetics of ART allocation in most condition-dependent polyphenic systems are not well studied apart from the bulb mite, Rhizoglyphus robini (Parrett et al., 2023, 2022; Radwan, 1995; see also Buzatto et al., 2012 for a further quantitative genetic example from a congeneric). In this species, morph allocation is a complex product of environmental (Smallegange, 2011) and genetic factors, with considerable additive genetic variance, which is likely in part linked to mutational load (Parrett et al., 2023).

Whether this is the case in other, similar ART systems is currently unknown. Studies of polyphenic beetles have yielded mixed results, with some finding of no significant heritability in morph allocation (O. taurus: Moczek & Emlen, 1999; Trypoxylus dichotomus: Karino et al., 2004). Both of these studies used quite small sample sizes, however (n = 17 and n = 18, respectively). By contrast, a selection study on Onthophagus acuminatus indicates that morph allocation is heritable (Emlen, 1996), and field observations of rapid evolution in morph allocation in O. taurus (Moczek et al., 2002) also indicate an important genetic component to morph allocation.

More widely, a heritable component to morph allocation linked to condition seems a reasonable assumption for most polyphenic systems. Condition dependence in sexually selected trait expression, as proposed by Rowe and Houle (1996), is widely thought to reflect mutational load and/or maladaptation (e.g., Dugand et al., 2019), and it seems unlikely that this will also not be captured in the discrete and nonlinear expression of sexually selected traits in species with polyphenic ARTs.

Here, we develop and analyze an individual-based model (IBM) of populations evolving in a changing environment, with males able to adopt either a competitive or a noncompetitive strategy on the basis of their condition—thus, we consider only the status-dependent case of ARTs with polyphenic males and not the very different frequency-dependent variant, nor do we consider sequential ARTs whereby tactics alter depending on age. IBMs are a useful approach to explore complex evolutionary processes, which interact with demography because they allow individual variation to be explicitly included (DeAngelis & Mooij, 2005).

We distinguish between "simultaneous" and "fixed" ARTs (Taborsky et al., 2008) where simultaneous ARTs can be expressed at the same time as the "normal" tactic by the same individual, as in three-spine sticklebacks where nest-holding males will also engage in sneak tactics (Rico et al., 1992), and fixed ARTs are those where one individual is only able to express one tactic, such as *O. taurus* and *R. robini* examples given above. Population persistence under directional environmental change and the speed of return to a near-optimal level of adaptation after a step change in the environment are both modeled.

### Methods

The model is coded in R (R Development Core Team, 2022). It simulates the dynamics and evolution of a spatially homogeneous, age-structured population of animals. The model proceeds as a series of discrete time steps of arbitrary length. All simulations described here were run for 500 time steps or until the population became extinct, whichever was shorter. Multiple simulations were run in parallel using the doparallel (Microsoft Corporation & Weston, 2022a) and foreach (Microsoft Corporation & Weston, 2022b) packages running in R version 4.3.0 on a 2021 MacBook Pro with a 10-core M1 Pro CPU and 32GB RAM.

The environment is described by a single continuous variable environment, which is best thought of as representing an environmental variable such as temperature or pH. The value for environ*ment* is initially set at 1, and two different treatments were used: directional change and step change. For directional change, the environment was held constant aside from a small amount of random variation for the first 25% of the time steps. Following this and for the rest of the simulation, the environment changes in a directional way, with a value drawn from a normal distribution with  $\mu$  equal to a parameter *directional rate* and  $\sigma$  = 0.005 added to the value at each time step. For the step-change treatment, the environment was allowed to remain constant aside from a small amount of random variation, but at time point 100, the value of environment had 0.33 added. This change is not sufficient to make the population become extinct, but populations exposed to such an environmental change will be some distance from being optimally adapted once the change has happened.

Populations are started with 200 individuals, which are allocated the following characteristics: sex (male or female, allocated at random); age (a value from 1 to 10, drawn at random from a uniform distribution); environmental genotype, which determines how well that individual is adapted to the environment (drawn for each individual from a normal distribution with  $\sigma = 0.25$ and  $\mu =$  the starting value for the "environment" variable); and resource, drawn at random from a uniform distribution with minimum 0 and maximum 1, which describes how each individual is able to acquire resources independently of the specific environmental effect being modeled. Including resource in the model here adds an element of environmental variance to the expression of the sexual trait, which in real systems will likely be determined by both environmental and genetic variance (Singh & Agrawal, 2022).

Each individual's condition is then calculated as:

Condition = resource - |environmentalgenotype - environment|

Individuals with condition < 0 die, so condition is a value between 1 and 0 for all living individuals.

The variable threshold determines whether a male with a given value for condition will develop into a major male (aggressive strategy) or a minor male (nonaggressive strategy). A fixed threshold for morph determination is consistent with the statusdependent model (Gross, 1996) and is consistent with what we know about morph allocation in animals such as the European earwig Forficula auricularia (Tomkins, 1999) and dung beetles from the genus Onthophagus (Emlen, 1997; Hunt & Simmons, 1997; Moczek & Emlen, 1999). As discussed in the Introduction section, there is uncertainty regarding the extent by which morph allocation reflects the genetic makeup of the males, and including the *resource* variable in the calculation means that there is a strong element of morph determination that is not a product of the individual's genotype. Simulations indicate that when

a population is well adapted (i.e., the population mean value for *environmental genotype* is equal to the *environment* variable), then *resource* explains about 10 times as much of the variation in morph as does the mismatch between genotype and environment. When the population is not well adapted (in this case with a difference of 0.4 between *environment* and *environmental genotype*), then *resource* explains about twice as much of the variance in morph than is explained by the mismatch between genotype and environment. We explore the impact of reducing the genetic contribution to morph allocation further in Supplementary Material.

In our model, the strategy a male follows is determined at age = 1 and then fixed for the lifetime of that individual, so this aspect of the model is representative of ARTs in examples such as bulb mites, dung beetles, and salmon where the alternative tactic is determined before sexual maturity. If *threshold* = 0, all males follow the aggressive strategy. Major males over the age of maturity (set at 2) express a display trait (*display*), which is equal to their *condition*. Minor males do not express such a trait. Two versions of the ART were used: one where the strategy is fixed (sensu Taborsky et al., 2008), and minor males do not attempt to compete for matings, and one where the strategy is simultaneous and minor males will compete for matings but will also try to acquire matings by other means.

Every time step, the *age* for each individual is incremented by 1. Following this, the probability of dying is calculated for each individual as:

$$p(death) = \frac{N_t}{K} + cost.display + 0.0154.age^2 + 0.169.age + 0.46$$

where  $N_t = population$  size, K determines the maximum population size, cost is the mortality cost of engaging in contests or energetically expensive displays, plus the costs of bearing the display trait. These are known to be substantial in many species (Clutton-Brock et al., 1997; Coulson et al., 2001; Liker & Székely, 2005; Loveridge et al., 2007; Promislow, 1997). Females and minor males do not express the display trait and do not pay this additional mortality cost: see Supplementary Material for a discussion of the effect of relaxing this such that major males pay a reduced or no cost. The quadratic component of the equation gives higher mortality for young and old individuals, with the coefficients having arbitrary values that gave an appropriate scale and shape of curve.

For mate allocation, males are divided into groups with size determined by a parameter *g*. If the ART is fixed, then only major males are allocated to these groups, if simultaneous then all males above the age of maturity are allocated to "mating groups" within each of which they compete for access to females for mating. The males in each group are ranked on the basis of the value of their display trait, and each female of reproductive age is allocated to one of the groups of males.

Each female will mate with one male from the group she is allocated to. The probability of mating is calculated for each male in a group on the basis of his ranking and the value of a parameter  $\beta$  as follows:

$$p(mate) = \frac{1}{rank^{\beta}} / \sum_{x=1}^{g} \frac{1}{x^{\beta}}$$

 $\beta$  controls the relationship between display trait expression and mating—this can be thought of as either the strength of female preference or the strength of the advantage in contests between males given by the display trait, with higher values of  $\beta$  meaning that males with large display traits have a greater advantage.

Condition is likely to have some effect on mating success in minor males, but this is likely to be a much weaker effect than in majors (e.g., Hunt & Simmons, 2001), and for simplicity, we assume that mating success in minor males is not associated with condition. Following the allocation of mates via contests between males, each female can also mate with a randomly selected minor male with a probability equal to the proportion of minor males multiplied by a parameter ART\_success, which allows the success of the minor strategists to be adjusted. Each female then produces a number of offspring equal to the maximum offspring multiplied by her condition, rounded to the nearest integer. For females who mate with both a major and a minor male, paternity for each offspring is allocated at random with equal probability for both males.

The environmental genotype is modeled as a quantitative, polygenic trait controlled by many alleles, so each offspring has an environmental genotype equal to the mean of its parents' environmental genotype value plus a value drawn from a normal distribution with mean = 0 and  $\sigma$  equal to a parameter called mutation.

For simulations run with directional change, the response variable used was whether the population became extinct over the course of the simulation. When the environmental change was a step change, the response variable used was the number of time units after the change before the upper quartile of the distribution of values of *environmental genotype* first became equal to or greater than the *environment* variable. This gives an indication of how long it takes the population to adapt to the new value of *environment*.

Supplementary Material contains output from simulations exploring the effects of reducing the additional mortality costs of expressing and signal, and of reducing the genetic contribution to morph allocation. It also has details of a further set of analyses exploring the effect of variables including the ART type, mating group size,  $\beta$ , and *ART\_success* in determining the Critical Rate of Environmental Change (Chevin et al., 2010) for these simulated populations.

# Results

### **Directional change**

Figure 1 shows the typical output from the model when the environment changes in a directional fashion for simulations with no alternative tactics expressed (Figure 1A-C), simultaneous alternative tactics (Figure 1D–F), and fixed alternative tactics (Figure 1G-I). While the environment remains roughly constant, the population fluctuates around the carrying capacity. There is some skew in the adult sex ratio as a consequence of the extra mortality cost experienced by major males, and when alternative tactics are expressed, the proportion of minor males in the population is roughly 40% (Figure 1F and I). At time = 125, the environment starts to undergo directional change. In each case, the population evolves in response, as can be seen by the increasing value of environmental genotype (the variable that determines the optimal value for the environment for each individual) in Figure 1B, E, and H. In all cases, the rate at which the population can adapt is slower than the rate of change of the environment and the median condition of the population declines (Figure 1C, F, and I), leading to both an increased death rate and lower reproduction by females. When alternative tactics are expressed, the proportion of minor males increases as a consequence of the generally lower condition of the population. In the case of simultaneous reproductive tactics, extinction occurs after roughly 350 time steps, but for the no alternative tactics and fixed alternative tactics examples the

population, while much reduced, persists until the end of the simulation. The gap between the median *environmental genotype* value and the median *environmental* value is an indication of the difference between population adaptation and the environment, and in Figure 1B and H, this gap can be seen to be less compared to Figure 1E, indicating that the population with simultaneous alternative tactics is not able to adapt as fast as the other two populations

Figure 2 shows the effect of alternative tactics on extinction probability for both fixed and simultaneous ARTs. In these simulations, the rate of directional change was set to a single value, which was sufficient to cause extinction in all simulations when mating was random-in this model, random mating occurs when either the mating group size or the threshold condition value (the value for condition which determines whether males follow the ART or not) for pursuing the alternative mating tactic threshold is 1. If there is relatively strong sexual selection and no males pursuing alternative strategies (mating tactic threshold = 0, mating group size > 1 and  $\beta$  > 1), then evolutionary rescue can occur, especially when the population is large and sexual selection is strong (mating group size is 10 and  $\beta$  = 2 or 4), in which case all populations are able to persist over the course of the simulation. The inclusion of some males pursuing simultaneous alternative tactics (left-hand column, mating tactic threshold = 0.25 or 0.5), however, negates this effect of sexual selection, and the probability of evolutionary rescue of these populations is reduced substantially.

By contrast with the situation for simultaneous alternative tactics, when the alternative tactics are fixed and minor males do not compete in groups for access to females, the presence of males pursuing alternative tactics increases the probability of evolutionary rescue. The right-hand column of Figure 2 shows the proportion of simulations where the population became extinct for a range of parameter values with fixed alternative tactics. When some lower condition males pursue the alternative tactic and do not compete with the other males, evolutionary rescue is more likely, occurring at lower values of  $\beta$  and mating group size than when males do not pursue the alternative tactics, and in smaller populations. In this case, the median value of *environmental genotype* can keep pace with the increasing value of *environment* and the population persists, albeit at a size far below the carrying capacity and with most individuals in poor condition.

### Environmental step change

Median return times for simulations that underwent an environmental step change are shown in Figure 3, with the return time here being the number of time steps after the step change before the upper quartile of the distribution of values for the environmental genotype became equal to or greater than the value for environment. The strength of sexual selection increases with both the mating group size and the parameter determining the strength of female choice or the advantage of higher quality males in contests within a group ( $\beta$ ) and the return time declines as both of these increase, indicating faster adaptation. When the alternative tactics are simultaneous, the presence of males following alternative tactics leads to slower return times and when males are more likely to follow the ART (threshold = 0.5 rather than 0.25), there is a greater increase in return time and therefore slower adaptation. When the ART is fixed, however, the inverse is found and the presence of males following the ART leads to shorter return times and therefore faster adaptation. Thus, consistent with the results from the directional change case above, simultaneous ARTs reduce adaptation, whereas fixed ARTs increase it.



**Figure 1.** Representative model output. (A–C) All males follow the "major strategy"—in other words, there are no alternative tactics. (D–F) Male alternative tactics are simultaneous. (G–I) Male alternative tactics are fixed. (A, D, G) Total population (green, upper line in all cases) plus the populations of mature females (purple, middle line) and males (orange, lower line). (B, E, H) The value of the environment variable at each time step (orange, upper line) plus the median and quartiles of the *environmental genotype* variable (green, lower line). (C, F, I) The median and quartiles of condition (green, line with associated paler lines for quartiles) plus the proportion of males adopting the minor strategy (orange, lower ine in C, F & I the line which is higher towards the right hand side of the plot). Parameter values for the simulations are K = 1,000, threshold = 0.25; max\_offspring = 6, group\_size = 6, ART\_success = 0.5, and  $\beta = 2$ . For the two cases where alternative tactics were available, threshold = 0.25; for the case when no alternative tactics were followed, threshold = 0.

When the mating tactic threshold = 1, then all males follow the ART, and so the return time is unaffected by the strength of sexual selection. Interestingly, when sexual selection is very weak or nonexistent (group size = 1), then the return times for the cases when all males follow the ART are actually lower than the alternatives. This is most probably because in this model males following the ART do not express the sexual signal and so do not pay a cost for doing so, whereas even when the group size = 1, males not following the ART will pay the cost for expressing the signal. Well-adapted, high-quality males will therefore have a somewhat longer life in the simulations where all males follow the ART and will consequently produce more offspring.

# Reproductive skew with simultaneous and fixed ARTs

Why do simulations with fixed ARTs show faster adaptation and an enhanced probability of evolutionary rescue? In this case, the low-condition males are excluded from the mating groups where males compete for matings. If all males are allocated to these groups, as when there are no males pursuing alternative tactics or when the alternative tactics are simultaneous, then purely by



**Figure 2.** Heatmap showing the probability of extinction under directional environmental change when alternative tactics are simultaneous (lefthand column) or fixed (right-hand column), calculated from 100 simulations running for 500 time steps for each combination of parameter values. The rate of directional change was set to 0.005 for all simulations. Other parameter values: max\_offspring = 6, ART\_success = 0.5, K = 500.

chance some groups will have no high-condition males, giving the low-condition males in those groups an opportunity to mate. If low-condition males do not compete, then all the groups of competing males will consist of high-condition males, enhancing the probability of evolutionary rescue because under these conditions all the females will mate with a relatively high-condition male, who most likely will have a genotype that is well matched to the changing environment.

To further explore this, we calculated the amount of reproductive skew in males for each time step for simulations with simultaneous and fixed male ARTs. Reproductive skew was calculated as the degree of skewness in the distribution of offspring per male using the moments package in R (Komsta & Novomestky, 2022). These data are shown in Figure 4. While the amount of reproductive skew is weakly negatively correlated with the proportion of minor males in the simultaneous case, when alternative tactics are fixed there is a clear positive relationship between reproductive skew and the proportion of males pursuing the alternative tactic.

## Discussion

ARTs are found across the animal kingdom, so if they have an impact on adaptation and evolutionary rescue, it is important that we understand this. On the basis of this model, we find that whether the ART is simultaneous or fixed is a crucial determinant of the population-wide evolutionary impact of the ART. When ARTs are simultaneous, the capacity of the population to adapt is reduced and evolutionary rescue is less likely when the environment is changing. When ARTs are fixed, however, and males pursuing the alternative tactic do not engage in contests with other males, the adaptive capacity of the population is enhanced and evolutionary rescue becomes more likely.

The reduced adaptive capacity of populations when the ART is simultaneous is intuitively understandable. Poorly adapted males, with a phenotypic optimum that is far from the existing value, will be in poor condition and unlikely to win contests with better-adapted males who are in better condition. By adopting the ART, these males will acquire matings that would otherwise be unavailable to them and will father offspring with maladapted alleles who would otherwise most likely be fathered by better-adapted males, inhibiting the process of adaptation at the population scale. These negative effects will likely be greatest in systems where females are unable to exert much choice over mating rates with males adopting the ART or where there is substantial conflict over matings. For example, male guppies (Poecilia reticulata) perform courtship displays in order to solicit matings with females. If, however, females are unreceptive, then males may adopt a coercive mating strategy in which females have little choice over and these coercive copulations have been shown to transfer a considerable amount of sperm (Pilastro & Bisazza, 1999). Although not modeled here, conflict between the sexes over matings, and more broadly sexual conflict, has been shown to have the potential to negate the positive populationlevel effects of sexual selection (Fricke & Arnqvist, 2007; Holland, 2002; Rundle et al., 2006), and this example adds to that potential.



**Figure 3.** Median return times for populations following an environmental step change. Return times are the number of time units before the upper quartile of the distribution of environmental genotype is equal to the value of *environment*. When the mating tactic threshold = 1, all males follow the ART, meaning that there is random mating in the simulation and males do not express sexual signal traits. Each value is the median from 100 simulations, and the carrying capacity was set at 500 for all simulations.

If the coercive ART imposes a cost to females via traits that increase male fitness at the expense of females, then again we would anticipate increased population-level negative effects when males adopt the ART, in a similar way to when harmful male traits are themselves condition dependent (Flintham et al., 2023).

The enhanced adaptive capacity of populations when the ART is fixed, on the other hand, does not seem intuitively obvious. As discussed in the results, this effect occurs because once the males following the ART are removed from competition, the average condition (and therefore degree of adaptation) of males undergoing competition will be higher, leading to greater reproductive skew toward the males in the best condition and consequently a greater representation of alleles conferring better adaptation in the next generation. This effect is robust and persists even when males pursuing ARTs have enhanced probabilities of acquiring matings and fathering offspring (see Supplementary Material).

One approach to thinking about this result is in the context of soft selection versus hard selection (Bell et al., 2021), whereby hard selection implies that an individual's fitness is largely determined by their phenotypic fit to environmental factors such as temperature or pH, whereas soft selection fitness is determined by an individual's social environment, meaning that it is their phenotype relative to their conspecifics that determines fitness. In general, males are thought to experience softer selection than females (Li & Holman, 2018), and this is the case in this model. One consequence of this soft selection on males is that a male with a relatively poor genotype in terms of phenotypic fit to the environment can still have high fitness if he is, by chance, in a group of competitors who all have even lower quality genotypes. Introducing the fixed ART hardens selection on males because the least well-adapted individuals are removed from competition, reducing the probability of a low-quality male achieving high fitness because of the chance composition of the group males he is competing with.

This surprising result for fixed ARTs is, of course, a theoretical outcome from a simulation model, and it is necessary to ask whether we should expect this mechanism to operate in real animal populations. There are no empirical studies that we are aware of that would provide a direct hypothesis test, but we can at least examine the assumptions of the model and consider whether there is any further evidence that might help us to address this.



**Figure 4.** Reproductive skew in males plotted against the proportion of minor males in the population for each time step of a simulation running for 500 time steps. (A) Alternative tactics are simultaneous, and male reproductive skew decreases slightly with the proportion of minor males. (B) Alternative tactics are fixed; male reproductive skew increases with the proportion of males following the minor strategy. Other parameter values: K = 500, ART\_success = 0.5, max\_offspring = 6, group\_size = 6, threshold = 0.25, and  $\beta = 2$ .

The strength of this effect relies on two assumptions in the model: First that the ART is fixed, and second that females will continue to mate with males that are not following the ART even when such males are rare. The first assumption that following the ART is fixed and that males pursuing this tactic do not engage in contests is certainly valid for many systems where ARTs are used-minor dung beetles do not invest in the horns that are necessary for contests with majors (Simmons et al., 2007), for example, and scrambler morphs of the mites Sancassiana berlesei and Rhizoqlyphus robini suffer high mortality from fights (Radwan, 1993; Radwan & Klimas, 2001) and therefore avoid fights with fighter morphs. The second assumption, that when males that are not following the ART are rare, all the females in the population will continue to mate with these males in the same way as when they are common, is weaker. Considering a female-choice system as an example, when such males are very rare some females are likely to mate with low-quality males if they are unable to easily find appropriate high-quality males. Even when females can locate suitable males the strength of female choice in a stressed system could be reduced by both a lack of prior experience (Bailey & Zuk, 2008; Macario et al., 2017) and the female's condition (Hunt et al., 2005). Similar arguments can be made about systems where the males engage in contests to monopolize females. In the extreme then, and in populations that are especially stressed by the changing environment the strength of the effect we have found is likely to be reduced, but it will not be eliminated altogether. In species where males compete to guard rare resources that are important for female reproduction, such as oviposition sites, then even when only a few high-quality males are present, these will still enjoy a high mating success because of the lack of rivals.

Does the phylogenetic distribution of ARTs tell us anything about whether they might enhance or reduce the persistence of a species? If the presence of males following ARTs reduces species persistence, we might expect to see a "twiggy" distribution of ARTs, with species expressing ARTs being found individually or in small clades at the tips of phylogenies, but with no large taxa expressing them, as has been suggested to be the case for asexual reproduction (Maynard, 1978; Schwander & Crespi, 2009). An analysis of the phylogenetic distribution of ARTs across the animal kingdom is not possible with current knowledge, but we can note that within the Coleoptera, there are a number of large clades with males that seem mostly to pursue ARTs. McCullough et al. (2015) examined the static allometry of the horns carried by males of 31 species from the Dynastinae (Hercules beetles) and found discontinuous relationships suggestive of ART expression in 30 of them. Within the Lucanidae (stag beetles), Matsumoto and Knell (2017) found evidence for complex polymorphisms, including species with three and even four male morphs, in all six species of Odontolabis examined. Within the Scarabaeinae (true dung beetles), the picture is somewhat more complex because, in the genus Onthophagus at least, male weaponry is known to be evolutionarily labile such that horns are rapidly gained and lost over evolutionary time (Emlen et al., 2005). Nonetheless, when the horned dung beetles are examined, the great majority of species seem to exhibit male polymorphism that is most probably associated with males following either a "guard" or a "sneak" strategy (Parrett et al., in prep.; Simmons et al., 2007). The existence of clades of animals where most or all males seem to pursue ARTs suggests that in the beetles at least the use of these tactics is not a significant contributor to the probability of extinction and is possibly enhancing species persistence. The behavior of the minor males has only been studied in detail in a relatively small number of species from the Scarabaeinae, but it is notable here that these minors all seem to follow something approximating to a fixed strategy-in O. acuminatus and O. taurus, e.g., minor males will enter burrows and remain there with females, but if challenged by a major male are unable to compete and resort to "sneak" behavior to acquire matings (Emlen, 1997; Moczek & Emlen, 2000). Future work might focus on ART distributions in fish: Both fixed and simultaneous ARTs are found within the ray-finned fish (Taborsky, 1998, 2008) and have been shown to have multiple independent origins (Mank & Avise, 2006). Addressing the distribution of each type of ART across this group may provide a direct test of the possibility that ART type influences extinction risk.

As a final point, we should point out that this model is only appropriate for considering the impact of some types of ART on the adaptive potential of a population. ARTs are diverse, and we have not considered, e.g., sequential ARTs where the reproductive tactic changes through the life of an animal. These are common in fish and mammals (Oliveira et al., 2008) and might have different effects on adaptive potential. ARTs that are based on genetic polymorphisms are very different from the polyphenic type modeled here, and whether these might alter adaptive potential is currently an open question.

## Supplementary material

Supplementary material is available online at Evolution Letters.

## Data and code availability

Model code, analysis code, and data from simulations are all available from Dryad at doi:10.5061/dryad.9zw3r22p1.

## Author contributions

Both authors conceived the study. R.K. wrote the model and analyzed the results. Both authors wrote the manuscript.

## Funding

R.K. is supported by Biotechnology and Biological Sciences Research Council (grant BB/V008110/1). J.P. is supported by the National Science Centre of Poland (grant 2020/39/D/NZ8/00069).

Conflict of interest: The authors declare no conflict of interest.

## Acknowledgments

The authors are grateful to Zac Wylde and two anonymous reviewers for their careful and constructive reviews, which led to a greatly improved manuscript.

## References

- Agrawal, A. F., & Whitlock, M. C. (2012). Mutation load: The fitness of individuals in populations where deleterious alleles are abundant. Annual Review of Ecology, Evolution, and Systematics, 43(1), 115– 135. https://doi.org/10.1146/annurev-ecolsys-110411-160257
- Bailey, N. W., & Zuk, M. (2008). Acoustic experience shapes female mate choice in field crickets. Proceedings of the Royal Society of London, Series B: Biological Sciences, 275(1651), 2645–2650. https:// doi.org/10.1098/rspb.2008.0859
- Bell, D. A., Kovach, R. P., Robinson, Z. L., Whiteley, A. R., & Reed, T. E. (2021). The ecological causes and consequences of hard and soft selection. *Ecology Letters*, 24(7), 1505–1521. https://doi. org/10.1111/ele.13754
- Bell, G. (2017). Evolutionary rescue. Annual Review of Ecology, Evolution, and Systematics, 48(1), 605–627. https://doi.org/10.1146/ annurev-ecolsys-110316-023011

- Buzatto, B. A., Simmons, L. W., & Tomkins, J. L. (2012). Genetic variation underlying the expression of a polyphenism. *Journal of Evolutionary Biology*, 25(4), 748–758. https://doi. org/10.1111/j.1420-9101.2012.02469.x
- Cally, J. G., Stuart-Fox, D., & Holman, L. (2019). Meta-analytic evidence that sexual selection improves population fitness. *Nature Communications*, 10(1), 2017. https://doi.org/10.1038/ s41467-019-10074-7
- Chevin, L. -M., Lande, R., & Mace, G. M. (2010). Adaptation, plasticity, and extinction in a changing environment: Towards a predictive theory. PLoS Biology, 8(4), e1000357. https://doi.org/10.1371/journal.pbio.1000357
- Clutton-Brock, T. H., Rose, K. E., & Guinness, F. E. (1997). Densityrelated changes in sexual selection in red deer. Proceedings of the Royal Society of London, Series B: Biological Sciences, 264(1387), 1509–1516. https://doi.org/10.1098/rspb.1997.0209
- Coulson, T., Catchpole, E. A., Albon, S. D., Morgan, B. J. T., Pemberton, J. M., Clutton-Brock, T. H., Crawley, M. J., & Grenfell, B. T. (2001). Age, sex, density, winter weather, and population crashes in soay sheep. Science, 292(5521), 1528–1531. https://doi.org/10.1126/ science.292.5521.1528
- DeAngelis, D. L., & Mooij, W. M. (2005). Individual-based modeling of ecological and evolutionary processes. Annual Review of Ecology, Evolution, and Systematics, 36(1), 147–168. https://doi.org/10.1146/ annurev.ecolsys.36.102003.152644
- Dougherty, L. R. (2021). Meta-analysis reveals that animal sexual signalling behaviour is honest and resource based. *Nature Ecology and Evolution*, 5(5), 688–699. https://doi.org/10.1038/ s41559-021-01409-z
- Dugand, R. J., Tomkins, J. L., & Kennington, W. J. (2019). Molecular evidence supports a genic capture resolution of the lek paradox. *Nature Communications*, 10(1), 1359. https://doi.org/10.1038/ s41467-019-09371-y
- Emlen, D. (1996). Artificial selection on horn length body size allometry in the horned beetle Onthophagus acuminatus (Coleoptera: Scarabaeidae). Evolution, 50, 1219–1230.
- Emlen, D. J. (1997). Diet alters male horn allometry in the beetle Onthophagus acuminatus (Coleoptera: Scarabaeidae). Proceedings of the Royal Society of London, Series B: Biological Sciences, 264(1381), 567–574. https://doi.org/10.1098/rspb.1997.0081
- Emlen, D. J., Marangelo, J., Ball, B., & Cunningham, C. W. (2005). Diversity in the weapons of sexual selection: Horn evolution in the beetle genus Onthophagus (Coleoptera: Scarabaeidae). Evolution, 59(5), 1060–1084.
- Engqvist, L., & Taborsky, M. (2016). The evolution of genetic and conditional alternative reproductive tactics. Proceedings of the Royal Society of London, Series B: Biological Sciences, 283(1825), 20152945. https://doi.org/10.1098/rspb.2015.2945
- Flintham, E. O., Savolainen, V., & Mullon, C. (2023). Male harm offsets the demographic benefits of good genes. Proceedings of the National Academy of Sciences of the United States of America, 120(10), e2211668120. https://doi.org/10.1073/pnas.2211668120
- Fricke, C., & Arnqvist, G. (2007). Rapid adaptation to a novel host in a seed beetle (*Callosobruchus maculatus*): The role of sexual selection. *Evolution*, 61(2), 440–454. https://doi. org/10.1111/j.1558-5646.2007.00038.x
- Godwin, J. L., Lumley, A. J., Michalczyk, L., Martin, O. Y., & Gage, M. J. G. (2020). Mating patterns influence vulnerability to the extinction vortex. Global Change Biology, 26(8), 4226–4239. https://doi. org/10.1111/gcb.15186
- Gross, M. R. (1996). Alternative reproductive strategies and tactics: Diversity within sexes. Trends in Ecology and Evolution, 11(2), 92– 98. https://doi.org/10.1016/0169-5347(96)81050-0

- Holland, B. (2002). Sexual selection fails to promote adaptation to a new environment. *Evolution*, 56(4), 721–730. https://doi. org/10.1111/j.0014-3820.2002.tb01383.x
- Hunt, J., Brooks, R., & Jennions, M. D. (2005). Female mate choice as a condition-dependent life-history trait. American Naturalist, 166(1), 79–92. https://doi.org/10.1086/430672
- Hunt, J., & Simmons, L. W. (1997). Patterns of fluctuating asymmetry in beetle horns: An experimental examination of the honest signalling hypothesis. *Behavior, Ecology and Sociobiology*, 41(2), 109–114. https://doi.org/10.1007/s002650050370
- Hunt, J., & Simmons, L. W. (2001). Status-dependent selection in the dimorphic beetle Onthophagus taurus. Proceedings of the Royal Society of London, Series B: Biological Sciences, 268(1484), 2409–2414. https://doi.org/10.1098/rspb.2001.1758
- Jarzebowska, M., & Radwan, J. (2010). Sexual selection counteracts extinction of small populations of the bulb mites. *Evolution*, 64(5), 1283–1289. https://doi.org/10.1111/j.1558-5646.2009.00905.x
- Karino, K., Seki, N., & Chiba, M. (2004). Larval nutritional environment determines adult size in Japanese horned beetles Allomyrina dichotoma. Ecological Research, 19(6), 663–668. https:// doi.org/10.1111/j.1440-1703.2004.00681.x
- Komsta, L., & Novomestky, F. (2022). Moments, cumulants, skewness, kurtosis and related tests. R package version 0.14.1. https:// CRAN.R-project.org/package=moments
- Lamichhaney, S., Fan, G., Widemo, F., Gunnarsson, U., Thalmann, D. S., Hoeppner, M. P., Kerje, S., Gustafson, U., Shi, C., Zhang, H., Chen, W., Liang, X., Huang, L., Wang, J., Liang, E., Wu, Q., Lee, S. M. -Y., Xu, X., Höglund, J., ... Andersson, L. (2016). Structural genomic changes underlie alternative reproductive strategies in the ruff (Philomachus pugnax). Nature Genetics, 48(1), 84–88. https://doi.org/10.1038/ng.3430
- Lank, D. B., Smith, C. M., Hanotte, O., Burke, T., & Cooke, F. (1995). Genetic polymorphism for alternative mating behaviour in lekking male ruff Philomachus pugnax. Nature, 378(6552), 59–62. https://doi.org/10.1038/378059a0
- Li, X. -Y., & Holman, L. (2018). Evolution of female choice under intralocus sexual conflict and genotype-by-environment interactions. Philosophical Transactions of the Royal Society B: Biological Sciences, 373(1757), 20170425. https://doi.org/10.1098/ rstb.2017.0425
- Liker, A., & Székely, T. (2005). Mortality costs of sexual selection and parental care in natural populations of birds. *Evolution*, 59(4), 890–897.
- Lorch, P. D., Proulx, S., Rowe, L., & Day, T. (2003). Condition-dependent sexual selection can accelerate adaptation. *Evolutionary Ecology Research*, 5, 867–881.
- Loveridge, A. J., Searle, A. W., Murindagomo, F., & Macdonald, D. W. (2007). The impact of sport-hunting on the population dynamics of an African lion population in a protected area. *Biological Conservation*, 134(4), 548–558. https://doi.org/10.1016/j. biocon.2006.09.010
- Macario, A., Croft, D. P., Endler, J. A., & Darden, S. K. (2017). Early social experience shapes female mate choice in guppies. *Behavioral Ecology*, 28(3), 833–843. https://doi.org/10.1093/beheco/arx043
- Mank, J. E., & Avise, J. C. (2006). Comparative phylogenetic analysis of male alternative reproductive tactics in ray-finned fishes. *Evolution*, 60(6), 1311–1316.
- Martínez-Ruiz, C., & Knell, R. J. (2017). Sexual selection can both increase and decrease extinction probability: Reconciling demographic and evolutionary factors. *Journal of Animal Ecology*, 86(1), 117–127. https://doi.org/10.1111/1365-2656.12601
- Matsumoto, K., & Knell, R. J. (2017). Diverse and complex male polymorphisms in Odontolabis stag beetles (Coleoptera:

Lucanidae). Scientific Reports, 7(1), 16733. https://doi.org/10.1038/ s41598-017-17115-5

Maynard, S. J. (1978). The evolution of sex. Cambridge University Press.

- McCullough, E. L., Ledger, K. J., O'Brien, D. M., & Emlen, D. J. (2015). Variation in the allometry of exaggerated rhinoceros beetle horns. Animal Behaviour, 109, 133–140. https://doi.org/10.1016/j. anbehav.2015.08.013
- Microsoft Corporation, & Weston, S. (2022a). doParallel: Foreach parallel adaptor for the "parallel" Package.
- Microsoft Corporation, & Weston, S. (2022b). foreach: Provides foreach looping construct. R package version 1.5.2. https://CRAN.Rproject.org/package=foreach
- Moczek, A., & Emlen, D. J. (2000). Male horn dimorphism in the scarab beetle, Onthophagus taurus: Do alternative reproductive tactics favour alternative phenotypes? Animal Behaviour, 59(2), 459–466. https://doi.org/10.1006/anbe.1999.1342
- Moczek, A. P., & Emlen, D. J. (1999). Proximate determination of male horn dimorphism in the beetle Onthophagus taurus (Coleoptera: Scarabaeidae). Journal of Evolutionary Biology, 12, 27–37. https:// doi.org/10.1046/j.1420-9101.1999.00004.x
- Moczek A. P., Hunt, L., Emlen D. J., Simmons L. W. (2002). Threshold evolution in exotic populations of a polyphenic beetle. Evolutionary Ecology Research, 4, 587–601.
- Neff, B. D., & Svensson, E. I. (2013). Polyandry and alternative mating tactics. Philosophical Transactions of the Royal Society of London, Series B: Biological Sciences, 368(1613), 20120045. https://doi. org/10.1098/rstb.2012.0045
- Oliveira, R. F., Taborsky, M., & Brockmann, J. H. (2008). Alternative reproductive tactics: An integrative approach. Cambridge University Press.
- Parrett, J. M., Chmielewski, S., Aydogdu, E., Łukasiewicz, A., Rombauts, S., Szubert-Kruszyńska, A., Babik, W., Konczal, M., & Radwan, J. (2022). Genomic evidence that a sexually selected trait captures genome-wide variation and facilitates the purging of genetic load. Nature Ecology & Evolution, 6(9), 1330–1342. https://doi. org/10.1038/s41559-022-01816-w
- Parrett, J. M., Chung, A. Y. C., Knell, R. J., & Slade, E. M. (in prep.). Idiosyncratic effects of anthropogenic alteration of tropical forest on the expression of sexually selected traits in dung beetles.
- Parrett, J. M., & Knell, R. J. (2018). The effect of sexual selection on adaptation and extinction under increasing temperatures. Proceedings of the Royal Society of London, Series B: Biological Sciences, 285(1877), 20180303. https://doi.org/10.1098/rspb.2018.0303
- Parrett, J. M., Łukasiewicz, A., Chmielewski, S., Szubert-Kruszyńska, A., Maurizio, P. L., Grieshop, K., & Radwan, J. (2023). A sexually selected male weapon characterized by strong additive genetic variance and no evidence for sexually antagonistic polyphenic maintenance. Evolution, 77(6), 1289–1302. https:// doi.org/10.1093/evolut/qpad039
- Parrett, J. M., Mann, D. J., Chung, A. Y. C., Slade, E. M., & Knell, R. J. (2019). Sexual selection predicts the persistence of populations within altered environments. *Ecology Letters*, 22(10), 1629–1637. https://doi.org/10.1111/ele.13358
- Pilastro, A., & Bisazza, A. (1999). Insemination efficiency of two alternative male mating tactics in the guppy Poecilia reticulata. Proceedings of the Royal Society of London Series B: Biological Sciences, 266(1431), 1887–1891. https://doi.org/10.1098/ rspb.1999.0862
- Promislow, D. E. L. (1997). Costs of sexual selection in natural populations of mammals. Proceedings of the Royal Society of London Series B: Biological Sciences, 247, 203–210.
- Radwan, J. (1993). The adaptive significance of male polymorphism in the acarid mite Caloglyphus berlesei. Behavior, Ecology and Sociobiology, 33(3), 201–208. https://doi.org/10.1007/bf00216601

- Radwan, J. (1995). Male morph determination in two species of acarid mites. *Heredity*, 74(6), 669–673. https://doi.org/10.1038/hdy.1995.91
- Radwan, J., & Klimas, M. (2001). Male dimorphism in the bulb mite, Rhizoglyphus robini: Fighters survive better. Ethology Ecology and Evolution, 13(1), 69–79. https://doi.org/10.1080/08927014.2001.9 522788
- Rico, C., Kuhnlein, U., & Fitzgerald, G. J. (1992). Male reproductive tactics in the threespine stickleback—An evaluation by DNA fingerprinting. *Molecular Ecology*, 1(2), 79–87. https://doi. org/10.1111/j.1365-294x.1992.tb00159.x
- Rowe, L., & Houle, D. (1996). The lek paradox and the capture of genetic variance by condition dependent traits. Proceedings of the Royal Society of London, Series B: Biological Sciences, 263, 1415–1421.
- Rundle, H. D., Chenoweth, S. F., & Blows, M. W. (2006). The roles of natural and sexual selection during adaptation to a novel environment. *Evolution*, 60(11), 2218–2225.
- Schwander, T., & Crespi, B. J. (2009). Twigs on the tree of life? Neutral and selective models for integrating macroevolutionary patterns with microevolutionary processes in the analysis of asexuality. *Molecular Ecology*, 18(1), 28–42. https://doi. org/10.1111/j.1365-294X.2008.03992.x
- Simmons, L. W., Emlen, D. J., & Tomkins, J. L. (2007). Sperm competition games between sneaks and guards: A comparative analysis using dimorphic male beetles. *Evolution*, 61(11), 2684–2692. https://doi.org/10.1111/j.1558-5646.2007.00243.x
- Sinervo, B., & Lively, C. M. (1996). The rock–paper–scissors game and the evolution of alternative male strategies. *Nature*, 380(6571), 240–243. https://doi.org/10.1038/380240a0
- Singh, A., & Agrawal, A. F. (2022). Sex-specific variance in fitness and the efficacy of selection. *American Naturalist*, 199(5), 587–602. https://doi.org/10.1086/719015
- Smallegange, I. M. (2011). Complex environmental effects on the expression of alternative reproductive phenotypes in the bulb

mite. Evolutionary Ecology, 25(4), 857–873. https://doi.org/10.1007/ s10682-010-9446-6

- Taborsky, M. (1998). Sperm competition in fish: 'Bourgeois' males and parasitic spawning. Trends in Ecology and Evolution, 13(6), 222–227. https://doi.org/10.1016/s0169-5347(97)01318-9
- Taborsky, M. (2008). Alternative reproductive tactics in fish. In R. F. Oliveira, M. Taborsky, & H. J. Brockmann (Eds.). Alternative reproductive tactics: An integrative approach (pp. 251–299). Cambridge University Press.
- Taborsky, M., Oliveira, R. F., & Brockmann, H. J. (2008). The evolution of alternative reproductive tactics: Concepts and questions. In R. F. Oliveira, M. Taborsky, & H. J. Brockmann (Eds.), Alternative reproductive tactics: An integrative approach (pp. 1–21). Cambridge University Press Cambridge.
- Tomkins, J. L. (1999). Environmental and genetic determinants of the male forceps length dimorphism in the European earwig Forficula auricularia L. Behavior, Ecology and Sociobiology, 47(1–2), 1–8. https://doi.org/10.1007/s002650050643
- Tomkins, J. L., & Hazel, W. (2007). The status of the conditional evolutionarily stable strategy. Trends in Ecology and Evolution, 22(10), 522–528. https://doi.org/10.1016/j.tree.2007.09.002
- Tomkins, J. L., Radwan, J., Kotiaho, J. S., & Tregenza, T. (2004). Genic capture and resolving the lek paradox. *Trends in Ecology and Evolution*, 19(6), 323–328. https://doi.org/10.1016/j. tree.2004.03.029
- R Development Core Team. (2022). R: A language and environment for statistical computing. R Foundation for Statistical Computing.
  R package version 1.0.17. https://CRAN.R-project.org/ package=doParallel
- Whitlock, M. C. (2000). Fixation of new alleles and the extinction of small populations: Drift load, beneficial alleles, and sexual selection. Evolution, 54(6), 1855–1861. https://doi. org/10.1111/j.0014-3820.2000.tb01232.x